Author’s response to reviews

Title: Modified constraint-induced movement therapy or bimanual occupational therapy following injection of Botulinum toxin-A to improve bimanual performance in young children with hemiplegic cerebral palsy: a randomised controlled trial methods paper

Authors:

Brian J Hoare (barneyhoare@hotmail.com)
Christine Imms (c.imms@latrobe.edu.au)
Hyam Barry Rawicki (barry.rawicki@southernhealth.org.au)
Leeanne Carey (lcarey@nsri.org.au)

Version: 8 Date: 5 July 2010

Author’s response to reviews:

Some minor typographical errors were corrected. This manuscript is not ready for publication.